Author’s response to reviews

Title: Biomarker Screening for Antenatal Depression in Women Who Underwent Caesarean Section: A Matched Observational Study with Plasma Lipidomics

Authors:
Zhuoxi Wu (WUSISSI527@163.com)
Peng Zhao (66949366@qq.com)
Zhonghong Long (FANNLZH@163.com)
Jie Li (824679901@qq.com)
Guiying Yang (472151685@qq.com)
Qingling Zhang (2805604446@qq.com)
Guangyou Duan (dgy1986anesthesia@126.com)
Hong Li (lh78553@163.com)

Version: 5 Date: 14 Aug 2019

Author’s response to reviews:

14th Aug. 2019
BMC Psychiatry

Dear editor

I along with my co-authors would like to re-submit the manuscript entitled “Biomarkers Screening for Antenatal Depression in Caesarean Women: A Matched Observational Study with Plasma lipidomics” to BMC Psychiatry. The previous manuscript ID is BPSY-D-18-00982R1.

Thank you again for the opportunity to revise the manuscript. We had revised the manuscript as required by the editor.
1. Authors' contributions
-- Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.
Respond: We have added relevant content to the text as required (Page 32, line 485).

2. Language certificate
-- Please remove any files from your file inventory that you do not wish to be published alongside your manuscript.
Respond: We have removed the language certificate as requested.

3. Clean manuscript
-- At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.
Respond: We have carefully rechecked the manuscript again and re-uploaded the clearer version of Figure 3. We hope that the revised manuscript is now suitable for publication in your journal. If there are any inappropriate places, please let us know and we will revise immediately.

I look forward to your reply.

Sincerely,

Hong Li, Director, Department of Anesthesiology, Xinqiao Hospital, Third Military Medical University, Chongqing, 400037, China.
Telephone: (+86)13608380123; Email: lh78553@163.com.